Author's response to reviews

Title: Evaluation of the toll-like receptor 6 Ser249Pro polymorphism in patients with asthma, atopic dermatitis and chronic obstructive pulmonary disease

Authors:

Sabine Hoffjan (sabine.hoffjan@rub.de)
Susanne Stemmler (susanne.stemmler@rub.de)
Qumar Parwez (elisabeth.petrasch-parwez@rub.de)
Elisabeth Petrasch-Parwez (elisabeth.petrasch-parwez@rub.de)
Umut Arinir (umut.arinir@rub.de)
Gernot Rhode (gernot.rohde@rub.de)
Karim Reinitz-Rademacher (karin.reinitz-rademacher@rub.de)
Gerhard Schultze-Werninghaus (gerhard.schultze-werninghaus@rub.de)
Albrecht Bufe (albrecht.bufe@rub.de)
Joerg T Epplen (joerg.t.epplen@rub.de)

Version: 3  Date: 2 September 2005

Author's response to reviews:

August 31, 2005

Editor-in-Chief
BMC Medical Genetics
BioMed Central Ltd
Middlesex House
34-42 Cleveland Street
London W1T 4LB, UK.

Dear Editor,

enclosed please find the manuscript, "Evaluation of the toll-like receptor 6 Ser249Pro polymorphism in patients with asthma, atopic dermatitis and chronic obstructive pulmonary disease" as a final manuscript for publication in BMC Medical Genetics.

We have made the requested formatting changes: the countries of all affiliated institutes were added, the keywords section removed from the manuscript and website addresses included in the reference list.

Further, we state now that the study was performed in compliance with the Helsinki declaration and that the authors have no competing interests.

Thank you in advance for publication of this manuscript.

Sincerely,

Sabine Hoffjan

- enclosures -